Title: Indacaterol improves lung hyperinflation and physical activity in patients with moderate chronic obstructive pulmonary disease - a randomized, multicenter, double-blind, placebo-controlled study

Authors:

Henrik Watz (h.watz@pulmoresearch.de)
Felix Krippner (felix.krippner@novartis.com)
Anne Kirsten (a.kirsten@pulmoresearch.de)
Helgo Magnussen (magnussen@pulmoresearch.de)
Claus Vogelmeier (claus.vogelmeier@med.uni-marburg.de)

Version: 2
Date: 15 July 2014

Author's response to reviews:

(1) Title Page: Please include the email addresses of all authors.
R: email addresses included

(2) Authors’ Contributions
R: section included

(3) Acknowledgements
R: section included

(4) Conclusion header
R: header included

(5) Cover Letter
R: Cover letter changed, sorry for the mistake. As indicated in the revised uploaded version, the ms has not been submitted anywhere else with the exception that some data were presented at the American Thoracic Society Congress in 2012, which is also referenced in the introduction.